**Author's response to reviews**

**Title:** Prevalence of risk factors for non-communicable diseases in the Mekong Delta, Vietnam: results from a STEPS survey

**Authors:**

- Luc H Pham (phluc534cairang@yahoo.com)
- Thuy B Au (bau@utas.edu.au)
- Leigh Blizzard (leigh.blizzard@utas.edu.au)
- Nhan B Truong (tbnhan56@yahoo.com)
- Michael D Schmidt (schmidtm@uga.edu)
- Robert H Granger (robert.granger@mac.com)
- Terence Dwyer (terry.dwyer@mcri.edu.au)

**Version:** 4  **Date:** 27 July 2009

**Author's response to reviews:** see over
27 July 2009

Dr. Melissa Norton
Editor-in-Chief
BMC Public Health

Dear Dr Norton,

Re: Submission of revised manuscript 1612781332645693

Thank you for your consideration of our manuscript entitled “Prevalence of risk factors for non-communicable diseases in the Mekong Delta, Vietnam: results from a STEPS survey”. We have made changes to the format of the manuscript as requested. I hereby submit the revised manuscript for consideration for publication in the BMC Public Health.

The corresponding author for this manuscript will be Associate Professor Leigh Blizzard, Tel: 61 3 6226 7719, Fax: 61 3 6226 7704, E-mail: Leigh.Blizzard@utas.edu.au. We hope that you find this revised manuscript to be suitable for publication.

Yours sincerely

Au Bich Thuy